Author’s response to reviews

Title: Development of a diabetes self-management +mHealth program: Tailoring the intervention for a pilot study in a low-income setting in Mexico

Authors:

Robin Whittemore (robin.whittemore@yale.edu)
Mireya Vilar-Compte (mireya.vilar@ibero.mx)
Soraya Méndez (soraya.burrola@ibero.mx)
Annel Lozano-Marrufo (annellozanomarrufo@gmail.com)
Roberta Delvy (roberta.delvy@yale.edu)
Mariana Pardo-Carrillo (marianapcarrillo@hotmail.com)
Selene De La Cerda (selene.delacerda@gmail.com)
Ninfá Pena-Purcell (NCPurcell@ag.tamu.edu)
Rafael Pérez-Escamilla (rafael.perez-escamilla@yale.edu)

Version: 2 Date: 21 Jan 2020

Author’s response to reviews:

We have reviewed all abbreviations in the manuscript to assure that the first time they are used, the full phrase is included. We have also provided a list of abbreviations at the end of the manuscript.

We have reviewed the publishing guidelines for authors.

We have revised this sentence to read more clearly. The revised sentence is: Adults who receive care in the Seguro Popular system have a mean age of 44.7 years, are 57% male, have low income (66.9% are in the 3 lowest socioeconomic quintiles), and 12.4% report a diagnosis of T2D.

- page 44 line 33 - instead of saying 'cellphone users age 25-44' it would be better just to say 'people aged 25-44'. Less repetitive and would flow better.

We agree and have made this change.
- page 19 line 27 - hypothesise rather than hypothesised? The earlier statement is in the present tense.

We agree and have made this change.

Lastly, we have changed the degree abbreviations for 2 of the authors to correctly reflect the title used in Mexico – MAN for Masters in Nutrition.